Author’s response to reviews

Title: Association of rs610604 in TNFAIP3 and rs17728338 in TNIP1 gene polymorphisms with psoriasis susceptibility: A meta-analysis of case-control studies

Authors:

Hai-bo Gong (gonghaibo2019@163.com)
Shu-tao Gao (gaoshutoo@126.com)
Xiong-ming Pu (Puxiongming@126.com)
Xiao-jing Kang (drkangxj666@163.com)
Xiu-juan Wu (xjwuxj@sina.com)

Version: 2 Date: 25 Apr 2020

Author’s response to reviews:

Reply to assistant Editor Comments:

1. Availability of data and materials
Reply:
We have revised these statements.

2. Authors’ Contributions
Reply:
a) We have amended the order of initials in regard to each author as appropriate.
b) We have made a clarification on the contributions of each author in the “Authors’ Contributions” part.

3. Overlap
Reply:
When we were writing the first version of this manuscript, we were doing another similar research about Behcet’s Disease. That makes the current manuscript contains some textual overlap with that one. We have reduced most of the overlap in this revised manuscript by using turnitin software.

4. Clean version
Reply:
We have uploaded a clean versions as required. This revised manuscript is a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

Reply to Reviewer reports:
Reply:
We have revised these details in this manuscript as the reviewer required.